Author’s response to reviews

Title: The effect of mobile personalised texting versus non-personalised texting on the caries risk of underprivileged adults: a randomised control trial

Authors:

Makiko Nishi (makikonishi@hotmail.com)

Virginia Kelleher (V.Kelleher@ucc.ie)

Michael Cronin (m.cronin@ucc.ie)

Finbarr Allen (f.allen@ucc.ie)

Version: 1 Date: 10 Nov 2018

November 10th, 2018

To
The Editors
BioMed Central Oral Health

Subject: Re-submission of “Original Research Article” titled “The effect of mobile personalised texting versus non-personalised texting on the caries risk of underprivileged adults: a randomised control trial”

Dear Sir/Madam

On behalf of all the authors, I would like to ask you to consider our revised manuscript entitled “The effect of mobile personalised texting versus non-personalised texting on the caries risk of underprivileged adults: a randomised control trial” for publication in BioMed Central Oral Health as an original research article. The reference number is OHEA-D-18-00185. Please find the attached point-by-point table summarising our reply to each comment of the reviewers and the corresponding changed parts where these can be viewed.

This manuscript has not been previously published and is not under consideration for publication elsewhere. However, as I informed on the last cover letter, the first manuscript before this revision formed part of my PhD project. My PhD was approved by University College Cork and the thesis is on line with a partial embargo. https://cora.ucc.ie/handle/10468/7001
Since I finished my PhD project in September 2018, my affiliation has been changed from University College Cork to a non-profit organisation, "Promoting Scientific Assessment in Prevention of Tooth Decay and Gum Disease".

All the authors have read the manuscript and have approved this submission.

Sincerely yours,

Makiko NISHI, B.Ed, D.D.S, M.D.P.H., Ph.D.
Chairperson of the Board
Non-profit Organisation
“Promoting Scientific Assessment in Prevention of Tooth Decay and Gum Disease”
Bunkyo-ku, Tokyo, Japan
Tel: +81-50-5308-4848
makikonishi@hotmail.com